Author's response to reviews

Title: PEARL study: Improvement of hospital care for patients with non-Hodgkin's lymphoma. Study protocol for a cluster randomized controlled trial.

Authors:

Jozette J.C. Stienen (J.Stienen@iq.umcn.nl)
Rosella P.M.G. Hermens (R.Hermens@iq.umcn.nl)
Lianne Wennekes (L.Wennekes@iq.umcn.nl)
Saskia A.M. van de Schans (sam_vd_schans@hotmail.com)
Heleen M. Dekker (H.Dekker@rad.umcn.nl)
Nicole M.A. Blijlevens (N.Blijlevens@hemat.umcn.nl)
Richard W.M. van der Maazen (R.vanderMaazen@rther.umcn.nl)
Eddy M.M. Adang (E.Adang@ebh.umcn.nl)
Johan H.J.M. van Krieken (J.vanKrieken@pathol.umcn.nl)
Petronella B. Ottevanger (P.Ottevanger@onco.umcn.nl)

Version: 4 Date: 24 June 2013

Author's response to reviews: see over
Nijmegen, June 24th, 2013

Subject: Manuscript re-submission, copy-edited version

Dear Editor,

Attached, you will find the carefully checked copy-edited version of our manuscript, titled ‘PEARL study: Improvement of hospital care for patients with non-Hodgkin’s lymphoma. Study protocol for a cluster randomized controlled trial’ for publication in Implementation Science.

I accepted all of the changes that were made to conform the manuscript with the journal style. As suggested in remark JA1, I reformulated the sentence marked (page 6). Furthermore, I have made one small change: I added the abbreviation cRCT (instead of RCT), since the protocol describes a cluster randomized controlled trial.

On behalf of all authors,
Yours sincerely,

Jozette Stienen, MSc

Radboud University Nijmegen Medical Centre
IQ healthcare, Internal code 114
PO Box, 9101
6500 HB Nijmegen, the Netherlands
0031 24 3667310 (phone)
0031 24 3540166 (fax)
J.Stienen@iq.umcn.nl